The Self-Limiting Dynamics of TGF-β Signaling *In Silico* and *In Vitro*, with Negative Feedback through PPM1A Upregulation



Junjie Wang^{1,2}, Lisa Tucker-Kellogg^{1,2,3}, Inn Chuan Ng⁴, Ruirui Jia⁵, P. S. Thiagarajan^{4,6}, Jacob K. White^{1,7}, Hanry Yu^{1,2,4,5,8}*

1 Computational and Systems Biology, Singapore-MIT Alliance, Singapore, **2** Mechanobiology Institute, Singapore, **3** Centre for Computational Biology, Duke-NUS Graduate Medical School, Singapore, **4** NUS Graduate School for Integrative Science and Engineering, National University of Singapore, Singapore, **5** Department of Physiology, National University of Singapore, Singapore, **6** School of Computing, National University of Singapore, **7** Department of Electrical Engineering and Computer Science, Massachusetts Institute of Technology, Cambridge, Massachusetts, United States of America, **8** Department of Biological Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts, United States of America

Abstract

The TGF- β /Smad signaling system decreases its activity through strong negative regulation. Several molecular mechanisms of negative regulation have been published, but the relative impact of each mechanism on the overall system is unknown. In this work, we used computational and experimental methods to assess multiple negative regulatory effects on Smad signaling in HaCaT cells. Previously reported negative regulatory effects were classified by time-scale: degradation of phosphorylated R-Smad and I-Smad-induced receptor degradation were slow-mode effects, and dephosphorylation of R-Smad was a fast-mode effect. We modeled combinations of these effects, but found no combination capable of explaining the observed dynamics of TGF- β /Smad signaling. We then proposed a negative feedback loop with upregulation of the phosphatase PPM1A. The resulting model was able to explain the dynamics of Smad signaling, under both short and long exposures to TGF- β . Consistent with this model, immuno-blots showed PPM1A levels to be significantly increased within 30 min after TGF- β stimulation. Lastly, our model was able to resolve an apparent contradiction in the published literature, concerning the dynamics of phosphorylated R-Smad degradation. We conclude that the dynamics of Smad negative regulation cannot be explained by the negative regulatory effects that had previously been modeled, and we provide evidence for a new negative feedback loop through PPM1A upregulation. This work shows that tight coupling of computational and experiments approaches can yield improved understanding of complex pathways.

Citation: Wang J, Tucker-Kellogg L, Ng IC, Jia R, Thiagarajan PS, et al. (2014) The Self-Limiting Dynamics of TGF-β Signaling *In Silico* and *In Vitro*, with Negative Feedback through PPM1A Upregulation. PLoS Comput Biol 10(6): e1003573. doi:10.1371/journal.pcbi.1003573

Editor: Christos A. Ouzounis, The Centre for Research and Technology, Hellas, Greece

Received January 15, 2013; Accepted February 27, 2014; Published June 5, 2014

Copyright: © 2014 Wang et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The work was supported by grants from the Singapore-MIT Alliance, Singapore-MIT Alliance for Research and Technology, Institute of Bioengineering and Nanotechnolgy, Mechanobiology Institute, and Janssen Cilag to HY; funding from the Singapore-MIT Alliance and the Mechanobiology Institute to LTK. JW is SMA (Singapore-MIT Alliance) scholar and ICN is NGS (NUS Graduate School for Integrative Sciences and Engineering) scholar. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: hanry_yu@nuhs.edu.sg

Introduction

Transforming Growth Factor- β (TGF- β), a regulator of cell migration and cell fate, is a pharmaceutical target for the treatment of metastatic cancer and fibrotic diseases [1]. Signal transduction from extracellular TGF- β to the cell nucleus through the Smad pathway is well documented [2-7]. The TGF- β ligand binds sequentially to the type II TGF- β receptor, a constitutively active kinase, and then to the type I receptor, to form a ligand-receptor complex (LRC). The type I receptor is activated by the type II receptor and then phosphorylates the R-Smads (Smad2 and Smad3) at two C-terminal serine residues. Upon phosphorylation, R-Smads form a homomeric complex or a heteromeric complex with Co-Smad (Smad4). The key outcome of the Smad cascade is the accumulation of phosphorylated R-Smad (phospho-R-Smad) in the nucleus, affecting the transcriptional regulation of many genes [7,8].

Smad signaling is known to decrease quickly after TGF-β stimulation, causing rapid decline of phospho-R-Smad after its initial peak. The HaCaT cell line has been adopted as an experimental model system for quantifying the detailed signaling of the TGF- β /Smad system. In HaCaT cells, there is a rapid decline of phospho-R-Smad after short exposure to TGF-β stimulation (30-45 min) [9,10], and a gradual decline after long exposure to TGF- β (6–24 hr) [9,11]. The duration of phospho-R-Smad activation could be crucial for regulation of different genes [12]. The self-limiting behavior of Smad signalling (i.e., negative regulation) may be caused by ligand-induced receptor inhibition [13-19], phospho-R-Smad dephosphorylation [9], phospho-R-Smad degradation [11,20-23], or other effects. Extensive experimental evidence has documented multiple modes of negative regulation, but the relative roles and combined effects are not well understood.

Previous computational models of TGF- β /Smad signaling have contributed important biological insights, but they have only

Author Summary

TGF- β signaling pathway regulates a variety of cellular responses, such as differentiation, migration and apoptosis. Phosphorylated R-Smad, the central signaling protein in this pathway, exhibits self-limiting behaviors: it not only decreases quickly after TGF- β is removed, but it also decreases slowly when TGF- β remains abundant. These two self-limiting behaviors are important to understand clearly because diseases such as cancer and fibrosis might benefit from treatments to decrease Smad signaling. Several negative regulatory effects have been reported previously, and we studied the dynamics of these effects with computational modeling. Analyzing the timing of negative regulation revealed that the three most widely accepted effects were not sufficient to explain the observed declines. After considering and excluding several alternative models, we arrived at a model in which TGF- β upregulated the phosphatase PPM1A. We tested for PPM1A upregulation in cell culture experiments. In addition, our model was able to explain why different durations of TGF- β exposure could cause seemingly opposite results about the importance of Smad degradation.

simulated some selected negative regulatory effects. Vilar et al. built a model of TGF-B receptor trafficking dynamics, including ligandinduced receptor degradation, which was able to simulate some key dynamic observations such as the peak and decline of phospho-R-Smad levels [24]. Models by Klipp and co-workers extended the work of Vilar et al. to include Smad phosphorylation and nuclear translocation [25], and to include transient versus sustained Smad signaling [12]. These models used simple representations for negative regulation, and gave a strong role to receptor degradation. The model by Schmierer et al. provided important insights into the Smad nucleo-cytoplasmic shuttling [10], but the only negative regulatory effect in this model was dephosphorylation. Other modeling studies have focused on robustness and in silico perturbation analysis [26,27]. Mathematical models have yielded important insights, but they have not represented TGF-B/Smad negative regulation with enough detail for analyzing the contributions of different negative regulatory effects, nor for evaluating alternative hypotheses.

In this work, we developed a series of computational models, representing individual and combination effects of R-Smad negative regulation. Comparisons between models and observations revealed negative regulation to occur at more than one timescale. We classified negative regulatory effects into fast-mode (5-240 min) and slow-mode (1-24 hr), depending on how quickly they act (and how quickly they equilibrate to steady state). Models then showed that at least one fast-mode and one slow-mode effect would be required for a model to fit the phospho-R-Smad dynamics in both short-exposure and long-exposure experiments. R-Smad Dephosphorylation was a fast-mode effect and it was strong enough to explain the fast-mode observations. Receptor Degradation and P-Smad Degradation were slow-mode effects, but they were too weak to explain the observed slow-mode decline. With a shortfall in explaining why R-Smad continues to decline hours after TGF- β stimulation, we sought a novel slow-mode effect. A second key finding of this work is a novel negative feedback effect, confirmed experimentally, in which the phosphatase PPM1A is upregulated after TGF- β stimulation. A final model hypothesizes how PPM1A might be upregulated with delayed activity, based on previously published molecular mechanisms for regulating PPM1A degradation [28,29]. Another final contribution we provide is an explanation for a previous controversy about proteasomal degradation of phospho-R-Smad [9,11,20,21]. Previous experiments inhibiting proteasomal degradation showed either strong effects [11,20,21] or no effects [9] on phospho-R-Smad levels. These seemingly contradictory trends were both mathematically consistent with our model, and the disparity could be explained by different durations of TGF- β exposure.

Results

A series of computational models were constructed (Table 1) examining negative regulatory effects in TGF-B/Smad signaling. All models share a common skeleton of Smad signaling (Figure 1), with TGF- β receptor internalization and trafficking, and Smad nucleo-cytoplasmic shuttling, based on previous models [10,24,25]. Molecular interactions were modeled using ordinary differential equations (ODEs) for mass action kinetics, and systems of ODEs were simulated using KroneckerBio [30] in Matlab (Mathworks, Natick, MA). The goodness-of-fit of parameter estimation was evaluated by the sum of squared errors (Text S7). The HaCaT cell line was used in our biological assays. Western blots and ELISA were used for time-series measurements of protein levels. Complete model specifications, parameter estimation, as well as biological experiments are specified in Materials and Methods and Supporting Information. In computational models and biological experiments, we used phosphorylated Smad2 (P-Smad2) to represent all phospho-R-Smad species, because P-Smad2 and P-Smad3 have highly similar dynamics in HaCaT cells [9].

Modeling the Negative Regulatory Effects

Smad signaling is enormously complex, as proven by a vast literature of previous work. We first considered negative regulatory effects from the published literature, for the purpose of selecting a set of effects relevant to our studies.

(1) RECEPTOR DEGRADATION. Smad7 (Inhibitor Smad or I-Smad) can target the ligand-receptor complex for degradation by recruiting E3-ligases [17-19]. In many cell types, Smad complexes can also induce the production of Smad7 (I-Smad), as a form of negative feedback [13,14]. However, in HaCaT cells, Smad7 levels are high and have minimal change after TGF- β stimulation [31]. Thus, we modeled RECEPTOR DEGRADATION as a first-order degradation reaction induced by a high constant level of Smad7. For comparison, we did try simulating a non-HaCaT model with Smad7 feedback, but the results were very similar to our model with a high constant level of Smad7 (data not shown). In addition to receptor degradation, there are two additional ways that I-Smad can antagonize the ligand-receptor complex: (a) It can block the activation site of the receptor kinase; and (b) It can recruit PP1c to dephosphorylate the type I receptor kinase [32]. We simulated these variants of the effect in Supporting Information (Text S2). In summary, we model RECEPTOR DEGRADATION as an intrinsically active process, but its target substrate is an activated species (the ligand-receptor complex), meaning that the reaction proceeds only when the target species is available due to TGF- β stimulation.

(2) **P-SMAD DEGRADATION.** R-Smad can be phosphorylated at its tail region or its linker region. We use the term "phospho-R-Smad" to refer to tail-region phosphorylation, which is more important for gene regulatory function [21] than the linker-region phosphorylation. Phosphorylation of nuclear R-Smad at its linker region causes Smurf2 to target R-Smad for proteasome-dependent degradation [11,21–23]. Assuming Smurf2 concentration to be

Table 1. A summary of the 8 models, showing the models with negative regulatory effects.								
Negative Regulatory Effect	Model	Number						
	-	2	m	4	ŝ	و	7	8
R-Smad Dephosphorylation	+	+			+	+	+	+
Receptor Degradation		+	+		+			,
P-SMAD DEGRADATION	,		,	+	+	+	+	+
ENDOGENOUS SYNTHESIS AND DEGRADATION OF R-SMAD	,		,	ı	ŗ		+	+
PPM1A Stabilization					,		,	+
Which dataset caused rejection of the model:	pS	T1R	pS	pS	tS	tS	pS	None
The + sign indicates that a negative regulatory effect is included in a model, and - indicates a negative regulatory effect is Nt pS: phospho-R-Smad time series. T1R: type I receptor time series. tS: total R-Smad time series. doi:10.1371/journal.pcbi.1003573.t001	IOT include	ġ						

Self-Limiting Dynamics of TGF-β Signaling

constant, and assuming linker-region phosphorylation to be proportional to tail-region phosphorylation [21], we set the rate of R-Smad degradation to be proportional to the concentration of nuclear R-Smad. Again, this is an intrinsically active process, but its target is an activated (phosphorylated) species, so the process only occurs when the target species has been activated by TGF- β stimulation.

(3) **R-SMAD DEPHOSPHORYLATION.** Phospho-R-Smad in the nucleus is dephosphorylated at the tail region specifically by PPM1A, a member of the PP2c family [9]. The rate of dephosphorylation is modeled to be proportional to the concentration of phospho-R-Smad in the nucleus. As previously, this is an intrinsically active process that only targets the activated species.

(4) Steady-state effects. Many factors have been found to regulate the gene expression, localization, degradation, or post-translational modification of Smad system proteins. A feedback loop is formed whenever these upstream factors are regulated by TGF- β . In contrast, if the upstream factors are independent of TGF- β (meaning constant during the course of a TGF- β -stimulation experiment), then we call them steady-state effects. Steady-state effects can alter the resting levels (the "initial concentrations") of the Smad system, and they can alter the absolute magnitude of phospho-R-Smad activation, but they do not explain the declining slope of phospho-R-Smad after TGF- β activation, because steady-state effects do not change during TGF- β stimulation. Because we study the decline rather than the absolute activation of phospho-R-Smad, we will not consider steady-state effects unless otherwise specified.

Any negative regulatory effect that was identified without TGF- β stimulation, we assumed to be a TGF- β -independent, steadystate effect. This assumption is not necessarily correct because biological networks often show greater interconnectivity than anticipated. Regulatory influences that we categorized as "steadystate effects" included degradation of Smad4 [33,34]; and sequestration of Smads [35–39].

(5) **PPMIA stabilization.** The PPM1A phosphatase can be protected from proteasomal degradation by binding PTEN [28], meaning that PTEN can serve as a negative regulator of Smad signaling. However, this effect had never been found to occur in response to TGF- β stimulation. In fact, the contrary was found. In the fibroblast cells where PTEN-mediated stabilization was observed, the stabilization was actually shut off by TGF- β stimulation [28], meaning that alteration of PPM1A stability contributed to self-perpetuation of TGF- β /Smad signaling, rather than providing a self-limiting mechanism. Although PTEN is a negative regulator of Smad signaling, it was found participating in a positive (double negative) feedback loop. Therefore, we assumed initially that altered stability of PPM1A was not contributing to the decline of Smad phosphorylation after TGF- β treatment.

(6) **Receptor internalization** was also included in all models, but we did not consider receptor internalization to be a negative effect, for saturating doses of TGF- β and for the 0.5–24 hr timescale. For additional explanation, see Text S1.

(7) Gene regulation and downstream effects. The gene regulatory functions of phospho-R-Smad are the result of complex interactions with many factors. Some proteins interfere with the interaction between the Smad complex and transcription factors, co-activators, or with the recruitment of HDAC [40,41]. These effects, although crucial for the gene regulatory functions of Smad signaling, are downstream of phospho-R-Smad. Our datasets measure phospho-R-Smad and our models study upstream events that regulate phospho-R-Smad levels. Thus, gene regulation and downstream effects are beyond the scope of our models.



Figure 1. The pathway diagram of Smad signaling (using symbols from BioCarta). The dashed arrows indicate those reactions which are modeled in black box. The red arrows indicate the negative regulatory effects: (1) PPM1A dephosphorylating phospho-R-Smad; (2) Smurf2 induced proteasome degradation of phospho-R-Smad; (3) I-Smad induced receptor inhibition; (4) PPM1A upregulation by Smad signaling. doi:10.1371/journal.pcbi.1003573.g001

(8) Sequestration of Smad by SnoN. R-Smad can be sequestered by cytoplasmic SnoN [42]. Although SnoN effects may be significant in many cell types, cytoplasmic SnoN is not detectable in HaCaT [43]. Therefore, we did not consider sequestration of Smad by SnoN in our modeling.

In summary, the most compelling negative regulatory effects in HaCaT appeared to be (1), (2), and (3) above, illustrated in

Figure 1. An additional self-limiting effect shown in Figure 1 is a negative feedback loop, described later, involving PPM1A upregulation.

Negative Regulation Occurs at Multiple Time Scales

R-Smad dynamics depend on the duration of TGF- β stimulation. When TGF- β is administered in excess (2 ng/ml) [44,45] for

24 hrs, phospho-R-Smad peaks at about 1 hr and then decays for 24 hrs [9]. When TGF- β is administered for 30 min and then removed (by washing following by receptor inhibition with the compound SB-431542), phospho-R-Smad is eliminated within 4 hrs [9,10].

Our first modeling studied the kinetics of the three negative regulatory effects selected from the literature review. We were curious whether they would have different kinetic implications for the system. Both short-exposure and long-exposure TGF- β treatment datasets (Figure 2) were utilized when building the models of negative regulation (Table 1, Table S2). The models were simulated to obtain the dynamics of their effects and to estimate their potential contributions to the down-regulation of phospho-R-Smad (0.5-24 hr).

Model 1, with R-SMAD DEPHOSPHORYLATION, was able to recapitulate the short-exposure TGF- β treatment experiment, as dephosphorylation is a fast process. This dephosphorylation model could turn off the signal once the stimulus was cut off (Figure 2A red curve), but it reached a steady state at about 1 hr and was not able to recapitulate the extended 24 hr decline of phospho-R-Smad in long-exposure TGF- β treatment (Figure 2A blue curve). Thus we describe R-SMAD DEPHOSPHORYLATION as a "fast-mode" effect. To explain the prolonged decline during long-exposure experiments, a complementary "slow-mode" might be provided by cumulative processes such as degradation. Model 2 combines RECEPTOR DEGRADATION and R-SMAD DEPHOSPHORYLATION. It succeeded in recapitulating the short-exposure TGF- β treatment very well, and it had moderately good agreement with the longexposure dataset (Figure 2B). As a control, we modeled RECEPTOR DEGRADATION alone (Model 3), but it could not provide an early decline in the short-exposure experiment (Figure 2C). Thus, RECEPTOR DEGRADATION serves as a slow-mode effect as it was able to explain the gradual and protracted decline of phospho-R-Smad in the long-exposure experiment but not the steep decline of phospho-R-Smad in the short-exposure experiment. Another cumulative process of decline is P-SMAD DEGRADATION. A model with P-SMAD DEGRADATION alone (Model 4) achieved significant negative regulation for the long-exposure case (Figure 2D), because P-SMAD DEGRADATION would persist for many hours. However, Model 4 had difficulty explaining both the shortexposure and long-exposure datasets simultaneously. If P-SMAD DEGRADATION is strong, it could recapitulate the steep decline after short-exposures, and if it is weak effect, it could recapitulate the gradual decline after long-exposures. Since it cannot be both strong and weak, it cannot explain both behaviors. Note that previous experimental evidence showed that P-SMAD DEGRADA-TION is not responsible for fast-mode effects in short-exposure conditions [9]. Having simulated each of the three negative regulatory effects in isolation, we could conclude that no single negative regulatory effect was able to explain phospho-R-Smad dynamics. We infer that the experimentally observed levels of phospho-R-Smad arise from a combination of fast-mode and slowmode effects (or from higher-order combinations of effects).

Many models have omitted P-SMAD DEGRADATION from simulations [10,12,24,25], perhaps because this effect was found to be insignificant in the experiments of Lin *et al.* [9]. Noting that the Lin experiments used short-exposure conditions, we asked whether P-SMAD DEGRADATION, a slow-mode effect, might have greater significance during the negative regulation induced by long-exposure treatments. Model 5 incorporated R-SMAD DEPHOS-PHORYLATION, RECEPTOR DEGRADATION, and P-SMAD DEGRADA-TION (Figure 2E). P-SMAD DEGRADATION was significant in this model (Figure 2F-G) when its impact was measured after more than 1 hr of TGF- β treatment. We also fitted a variety of models to the short-exposure and long-exposure experiments. The cumulative difference in phospho-R-Smad between +MG132 and -MG132 was minor in the short-exposure experiment and significant in the long-exposure experiment (Figure 2H). As yet, we have no basis for knowing which type of slow-mode degradation would be most important in R-Smad signaling.

Receptor Degradation Is Not Supported by Experiments

We next tried to assess the relative impact of two slow-mode effects, RECEPTOR DEGRADATION and P-SMAD DEGRADATION, on the dynamics of phospho-R-Smad in long-exposure TGF-B treatment. The rate constant for RECEPTOR DEGRADATION and the rate constant for P-SMAD DEGRADATION were varied in silico (Figure 3A), showing that many ratios were equally good at fitting the observed dynamics. Several of the successful models exhibited a strong decline in T1R, the type I receptor (Figure 3B). Moreover, the degree of T1R decline was correlated with the rate of RECEPTOR DEGRADATION and the rate of P-SMAD DEGRADATION (Figure 3C). Thus, to quantify the relative contribution of RECEPTOR DEGRADATION and P-SMAD DEGRADATION in HaCaT cells, we measured T1R experimentally at 9 time points (from 15 min to 24 hr) after TGF- β stimulation (with n = 3 replications and significance determined by Student's t-test). Surprisingly, there was no significant loss of T1R (type-I receptor) observed in experiments (Figure 3D-E), even at late time points. (As positive control, phospho-R-Smad time series concentrations were measured in Figure 4C). Previous work has already shown that T2R (type II receptor) shows no decrease after 2 ng/ml of TGF-β treatment in HaCaT cells [46]. Unchanged receptor levels indicate that RECEPTOR DEGRADATION is very weak in HaCaT cells. A weak role for RECEPTOR DEGRADATION has also been suggested by the experimental work of Clarke et al. [47]. Other forms of receptor inactivation or sequestration may occur without changing the total T1R concentration, but there is less published evidence for these possibilities (modeling analysis rejected these possibilities as well, in Text S2). Note that the set of models (Figure 3A) capable of explaining the dynamics of phospho-R-Smad decline all exhibited a negative correlation between the degree of RECEPTOR DEGRADATION and the degree of P-SMAD DEGRADATION (Figure 3F), suggesting that these two effects would be balanced alternatives. In light of our experimental finding that RECEPTOR DEGRADATION is a very weak effect, we next turned to P-SMAD DEGRADATION as the alternative slow-mode effect to explain the long-term decline of phospho-R-Smad.

P-SMAD DEGRADATION IS NOT Sufficient to Explain the Peak and Decline of Phosphorylated R-Smad

A model with R-SMAD DEPHOSPHORYLATION and P-SMAD DEGRADATION (Model 6, without RECEPTOR DEGRADATION) provided an excellent fit to both the long-exposure and short-exposure treatment data (Figure 4A). However, an unavoidable consequence of this model was dramatic decline of total R-Smad (Figure 4B). Previous experiments in HaCaT cells failed to observe a large fold-change of total R-Smad [9] but the amount of decline was not quantified. To clarify this potential conflict, we repeated the experimental measurement of total R-Smad levels after TGF-β treatment, using ELISA assays, a more quantitative method. Measurements of total R-Smad at 7 time points during 24 hrs of TGF-B treatment showed no significant decrease of total R-Smad (Figure 4B–C). There is an apparent conflict between the constant level of total R-Smad (observed experimentally) and the significant degradation of R-Smad induced by TGF-B (according to Model 6).



Figure 2. Model fitting results with different combinations of known negative regulatory effects. (A–E) Dots: experimental data from Lin *et al.* [9]. All P-Smad2 measurements used total cell lysate. Curves: the model simulations were fitted to the two sets of data simultaneously. (A) Model 1: R-Smad DephosphoryLation; (B) Model 2: R-Smad DephosphoryLation and Receptor Degradation; (C) Model 3: Receptor Degradation; (D) Model 4: P-Smad Degradation; (E) Model 5: R-Smad DephosphoryLation, Receptor Degradation and P-Smad Degradation. The reactions of each model are listed in the Supporting Information. (F–H) Predictions of the best-fit model (Model 5) in MG132 pre-treated cells. Simulation of MG132 treatment was performed by turning off the Smurf2-induced P-Smad Degradation (setting $kdeg_{pSmad2} = 0$) in Model 5. (F) Comparison of the model prediction and experimental data from Lin *et al.* [9] in the short-exposure experiment. (G) Model prediction in the long-exposure experiment. The green shaded area shows the

cumulative difference between +MG132 and -MG132. (H) A histogram plots the cumulative differences seen in the short-exposure experiment (red) and the long-exposure experiment (blue). doi:10.1371/journal.pcbi.1003573.g002

Degradation might be more difficult to rule out if we consider TGF- β -stimulated degradation in combination with **ENDOGENOUS SYNTHESIS AND DEGRADATION OF R-SMAD**. If endogenous R-Smad is synthesized in an unphosphorylated form, and targeted by

Smurf2 for degradation only in its phosphorylated form, then can P-SMAD DEGRADATION explain the decline of phospho-R-Smad despite the constant levels of total R-Smad? We therefore expanded the model to include ENDOGENOUS SYNTHESIS AND



Figure 3. Predictions and validations of Receptor Degradation. (**A**) Different rates of I-Smad-induced Receptor Degradation ($klid = 10^{-6} \sim 10^{-2}$) were applied to Model 5, and the rate of Smurf-induced P-Smad Degradation ($kleg_{pSmad2}$) was fitted to the short-exposure experimental data (red dots) and the long-exposure experimental data (blue dots). All the other parameters were kept the same as those in Model 5 (**B**) Different Receptor Degradation rates led to different levels of the type I receptor (T1R). Green curves were generated from all models in panel (A) with $klid = 10^{-6} \sim 10^{-2}$ and $kdeg_{pSmad2}$ estimated. (**C**) In the fitted models in panel (A), the T1R level has negative correlation with the Receptor Degradation rate (klid) but positive correlation with the P-Smad Degradation rate ($kdeg_{pSmad2}$). (**D**) Quantified data from 3 replicates of the western blot in (E). There is no significant loss of the T1R comparing the first and last data points (P>0.05). (**E**) Western blot of the T1R from whole cell lysates of HaCaT cells treated with TGF- β for 24 hrs (representative of 3 replicates). (**F**) In the fitted models in panel (A), the rates of Receptor Degradation (klid) and P-Smad Degradation ($kleg_{pSmad2}$) have negative correlation. doi:10.1371/journal.pcbi.1003573.g003



Figure 4. Simulations and experiments for P-SMAD DEGRADATION. (A) Model 6 with P-SMAD DEGRADATION and R-SMAD DEPHOSPHORYLATION (but no RECEPTOR DEGRADATION) was fitted to both the short-exposure (red) and long-exposure (blue) experimental data. (**B**) Model 6 predicted significant loss of total R-Smad (green curve), while ELISA measurements showed insignificant change (P>0.05, comparing the first and last data points) in total R-Smad concentration (green dots). (**C**) ELISA measurements of phospho-R-Smad are consistent with previous measurements performed by Western blot [9]. Cell lysates were from the same samples as panel B. (**D**) Model 7 was fitted to the phospho-R-Smad data while constraining the total R-Smad level to be constant.

doi:10.1371/journal.pcbi.1003573.g004

DEGRADATION OF R-SMAD (Model 7). However, Model 7 diverged strongly from the observed dynamics of phospho-R-Smad, when constrained to maintain a constant level of total R-Smad.

To summarize these results, P-SMAD DEGRADATION can only affect the shape of the phospho-R-Smad curve if it is not balanced by synthesis, in which case it would cause an unrealistic decline in the total Smad levels. If P-SMAD DEGRADATION is balanced by Smad synthesis, then it can only affect the height but not the shape of the phospho-R-Smad curve. Therefore we can rule out strong P-SMAD DEGRADATION (not balanced by synthesis) as an explanation for the later decline in the phospho-R-Smad curve shape. We cannot rule out the presence of significant P-SMAD DEGRADATION accompanied by Smad synthesis.

Hence, our model-driven experimental tests, sensitivity analysis (Text S3), and modeling analysis showed that P-SMAD DEGRADA-TION and RECEPTOR DEGRADATION were not sufficient to explain the 1–24 hr decline in phospho-R-Smad dynamics. We next sought some other negative regulatory effect that could help explain the peak and decline of phospho-R-Smad after a long exposure to TGF- β .

PPM1A Is Upregulated after Treatment with TGF- β

After excluding the three well-accepted effects of Smad negative regulation, we then examined possible alternative influences at different steps along the Smad pathway, seeking quantitative consistency with the observed peak and decline of phospho-R- Smad. One scenario that could not be rejected on kinetic grounds was upregulation of PPM1A, the phosphatase targeting phospho-R-Smad. If PPM1A were to be upregulated by TGF- β signaling, this could help explain the decline of phospho-R-Smad after long exposure to TGF- β (Text S4). To test this possibility, we performed Western blots of the PPM1A protein after TGF- β treatment. HaCaT cells were treated with 2 ng/ml of TGF- β and measured after 0.25, 0.5, 1, 2, 4, or 8 hr. We found that the intensity of the PPM1A western blot band increased 2.4-fold after 1 hour of TGF- β treatment (p<0.05, Figure 5A-B). To the best of our knowledge, this is the first study to report that TGF- β causes upregulation of the PPM1A phosphatase.

The increased abundance of PPM1A after TGF- β stimulation could be due to some type of decreased degradation and/or increased production. To aid future studies in investigating how the upregulation occurs, we have constructed a hypothetical mechanism, PPM1A STABILIZATION, in which we speculate that PTEN may be involved. Model 8 includes PPM1A STABILIZATION plus all the mechanisms of Model 7 (R-SMAD DEPHOSPHORYLATION, P-SMAD DEGRADATION, and ENDOGENOUS SYNTHESIS AND DEGRA-DATION OF R-SMAD). Text S5 provides a full specification of Model 8.

In previous studies, Bu et al. found that PTEN can bind to PPM1A and protect it from degradation [28]. These studies of PPM1A stability occurred in fibroblasts, where TGF- β caused dissociation of PTEN and PPM1A, leading to downregulation, not upregulation of PPM1A. In other words, they found PTEN to be a



Figure 5. Predictions and validation of PPM1A UPREGULATION. (A) Western blot of PPM1A in HaCaT cells with 2 ng/ml TGF- β treatment, representative of 3 replicates. (B) Model 8 predicted PPM1A upregulation under long-exposure of TGF- β (green curve). Our experimental validation showed significant upregulation of PPM1A (green dots, quantification from 3 Western blots, P<0.05 comparing the untreated data point and the 1 hr data point). (C) Model 8 was fitted to the long-exposure and the short-exposure phospho-R-Smad experimental data. (D) Model 8 predicted unchanged T1R levels (green curve), in agreement with our experimental results (green dots). (E) Model 8 predicted unchanged total R-Smad levels (green curve), in agreement with our experimental results (green dots). (F) Red solid curve shows simulation of Model 8 with short-exposure (30 min) of TGF- β , while the yellow dotted curve shows the same simulation except with MG132 pre-treatment. MG132 was simulated as turning off

Smurf2-induced P-SMAD DEGRADATION ($kdeg_{pSmad2} = 0$), but having no impact on basal degradation of Receptors, unphosphorylated R-SMAD, or PPM1A. (G) The blue solid curve shows simulation of Model 8 with long-exposure (8 hr) of TGF- β , and the green dotted curve shows the same simulated except with MG132 pre-treatment. (H) The relative change in P-Smad2 levels after MG132 treatment, calculated from Eq. 1 and simulations of Model 8. The P-Smad2 change simulated using Model 8 in both short-exposure (30 min, red curve) and long-exposure (8 hr, blue curve) simulations was compared with the P-Smad2 changes in the experimental results of Lin *et al.* [9] (30 min-exposure, red dots) and Alarcon *et al.* [21] (6 hr-exposure, blue dots). Data points from Alarcon *et al.* [21] were quantified from one published image. The discrepancy between our simulations and Alarcon *et al.* for the 7 hr measurement may be partially explained by MG132-independent differences. Their -MG132 control decreases much faster than that from Lin *et al.* [9] and from our experiments. doi:10.1371/journal.pcbi.1003573.g005

negative regulator of Smad signaling, but in their fibroblasts, TGF- β decreased this negative effect causing self-perpetuation (positive feedback) rather than self-limitation (negative feedback) of the Smad signal. The binding of PTEN in response to TGF- β is known to differ between fibroblasts and HaCaT keratinocytes. Hjelmeland et al. found that in HaCaT cells, TGF-ß stimulation caused formation of a PTEN-Smad complex [29], not dissociation of the PTEN complex [28]. They did not measure participation of PPM1A in that complex, but based on our analysis of the trends from [28] and [29], we propose a scaffolding role for phospho-R-Smad to promote association between PTEN and PPM1A in HaCaT cells. In other words, Model 8 speculates that TGF-B stimulation would induce PTEN association to stabilize PPM1A. This implies that there is some new or unknown mechanism upstream of PTEN, to explain why TGF-B signaling would promote PTEN-PPM1A association in one cell type and dissociation in another cell type. Model 8 assumes that PTEN and PPM1A would have a low on-rate for binding each other in HaCaT cells without phospho-R-Smad, but they would readily form a ternary complex in the presence of phospho-R-Smad. Thus, PPM1A would not be strongly stabilized in unstimulated HaCaT cells. After TGF-B stimulation, the phospho-R-Smad mediated association between PTEN and PPM1A would protect PPM1A from degradation and create negative feedback in the system. Note that Model 8 does not imply any alteration of total PTEN protein levels, merely the recruitment of PTEN by phospho-R-Smad into complexes with PPM1A. Simulations of Model 8 in Supplementary Figure S9 confirm that total PTEN levels could in theory remain constant (as observed in [29]) while levels of the PTEN-PPM1A complex could change over time.

Simulations of Model 8 were consistent with all the observed dynamics for the impact of TGF- β on HaCaT cells. This model was sufficient to explain the complete dynamics of phospho-R-Smad after short or long exposures to TGF- β (Figure 5C), the dynamics of PPM1A (Figure 5B), and the unchanged levels of T1R and total R-Smad (Figure 5D-E). With the key experimental trends satisfied, we next tested Model 8 against another dataset, obtained from combination treatment with TGF- β and a chemical inhibitor MG132.

Previous studies assessed P-SMAD DEGRADATION using MG132 to inhibit proteasomal degradation, but with conflicting conclusions: Massague et al. saw a strong impact, implying an important role for degradation [11,21], while Lin *et al.* found negligible impact from MG132 [9]. Both protocols measured the long-term dynamics of phosphorylated Smad2, but the Lin protocol triggered phosphorylated Smad2 using a 30 min exposure to TGF- β , while the Massague protocol used a 6 h exposure. Simulations of Model 8 with MG132 inhibition of proteasomal degradation show that MG132 would have minimal impact on Smad signaling, when triggered by short exposure to TGF-B (Figure 5F). In surprising contrast, MG132 would have a strong impact on Smad signaling, when phospho-R-Smad is triggered by longer exposures to TGF- β (Figure 5G). Figure 5H compares the P-Smad2 Change calculated from Figure 5F (red curve) and Figure 5G (blue curve) with experimental data from Lin et al. (Figure 1C in [9]) (red dots) and Alarcon *et al.* (Figure 2G in [21]) (blue dots). The P-Smad2 Change was calculated as Eq. 1.

$$P - Smad2 Change_{t=t_i}$$

$$= \frac{[pSmad2^{-MG132}_{total}]_{t=t_i} - [pSmad2^{+MG132}_{total}]_{t=t_i}}{[pSmad2^{-MG132}_{total}]_{max}}$$
(Eq.1)

Model 8 shows, mathematically, that the Lin observations and the Massague observations can be generated from the same system. Model 8 contains hypothetical mechanisms (e.g., PPM1A STABILIZATION) and imperfect parameters (e.g., reaction rate constants), but it suffices to prove that the seeming conflict between Lin et al. and Massague et al. is not necessarily a contradiction. In summary, the combination of several negative regulatory effects was consistent with, and sufficient to explain, the observed nuances of negative regulation and degradation in the Smad signaling system.

Discussion

Several negative regulatory effects in the Smad signaling pathway have been identified and individually studied [9,11,13,14,17,21,32,48]. We focused our modeling and experiments on these specific effects with published evidence. R-SMAD DEPHOSPHORYLATION by PPM1A is widely recognized to be a strong form of negative regulation, having significant fast-mode impact. However, the known slow-mode effects could only recapitulate phospho-R-Smad dynamics at the expense of very strong, cumulative degradation; as much as 90% decrease of T1R at 24 hr (Figure 3B), or 90% decrease in total R-Smad at 24 hr (Figure 4B). Our experimental measurements in HaCaT found that total T1R protein levels did not decline significantly (Figure 3B, 3D), nor did total R-Smad (Figure 4B). This contrasts with previous work in 293T and COS-1 cells [17,19]. In [17], 293T cells were transfected with I-Smad which was able to induce significant receptor degradation. The significant degradation seen in [17] may be due to transfection [47] or may be due to cell line differences. Although most dynamic models of signal transduction represent an amalgam of findings from multiple cell lines, our model (and the previous models we rely on) are specific to the HaCaT cell line. Thus a discrepancy with [17] is not necessarily a flaw of our model.

In light of our experimental measurement that TGF- β treatment does not cause any significant drop in total R-Smad levels, and the evidence showing no significant decline in type I or type II receptor levels, we conclude that degradation effects, if they occur in HaCaT, must be counterbalanced by endogenous synthesis. Model 7 simulated a balance of synthesis and degradation (ENDOGENOUS SYNTHESIS AND DEGRADATION OF R-SMAD) such that phospho-R-Smad was degraded while unphosphorylated Smad was synthesized; this model was not able to induce the observed decline of phospho-R-Smad in long-exposure experiments. The first key contribution of our work was to

conclude that degradation of R-Smad or T1R, with or without endogenous synthesis, is not sufficient to explain the slow-mode of Smad negative regulation in HaCaT cells. Degradation with synthesis remains a plausible effect of negative regulation, but it must occur alongside other effects. Figure 6 shows the relative contributions of different negative regulatory effects in our final model (Model 8): R-SMAD DEPHOSPHORYLATION was crucial for maintaining a limited level of phospho-R-Smad (compare red versus yellow curves); PPM1A STABILIZATION was capable of explaining the decline after the peak of phospho-R-Smad (compare yellow versus green curves); and P-SMAD DEGRADATION could further adjust the absolute level of phospho-R-Smad (compare green versus blue curves).

The second key contribution of our work was the discovery of a novel feedback loop in which the PPM1A protein is significantly upregulated after TGF-B treatment. Feedback loops have crucial importance in dynamical systems because they create nonlinear responses and permit self-regulation (by converting a directed subgraph into a connected subgraph). In our study, the new feedback loop via PPM1A was significant enough to allow the model to finally explain the observed trends of phospho-R-Smad decline after TGF-B treatment (Figure 6, yellow versus green curves). Because PTEN is known to stabilize PPM1A against degradation [28], we built a model to illustrate hypothetical dynamics of PTEN-induced PPM1A sequestration, including delayed enzymatic activity for PPM1A. Note that in previous experiments, the influence of PTEN served as a positive feedback loop (PTEN-induced stabilization was inhibited by TGF-β [28]), not negative feedback. HaCaT cells are an accepted model system for understanding how epithelial cells respond to TGF-β; and it will be interesting for future work to test which cell types utilize PPM1A regulation for negative feedback.

Model 8 shows that PPM1A STABILIZATION, with delayed nuclear import, was sufficient to reconcile the early upregulation of PPM1A total protein with later decline of phospho-R-Smad. Our theoretical model could be useful for the design of experiments to determine how the upregulation actually occurs. Future work should test whether PTEN stabilizes and/or sequesters PPM1A in HaCaT after TGF-β treatment, as illustrated in Model 8. Our model would recommend testing for



Figure 6. Relative contributions of different negative regulatory effects in our final model (Model 8). The effects of R-SMAD DEPHOSPHORYLATION, PPM1A STABILIZATION and P-SMAD DEGRADATION were removed one after another from Model 8, and the dynamics of P-Smad2 were simulated after 24 hr TGF- β treatment. In the absence of any negative regulatory effect (red curve), P-Smad2 levels climbed rapidly beyond the scale of the plot; and are not shown for time points after 1 hr.

doi:10.1371/journal.pcbi.1003573.g006

PPM1A-PTEN binding at 30 min–1 hr to catch the peak interaction, but testing for increased PPM1A activity at 4 hr, significantly later than the upregulation.

Careful examination of a broader set of previous work reveals some issues that appear to be discrepancies. The steepness of phospho-R-Smad decline in HaCaT appears to differ slightly between the experiments of Massague and colleagues in [11,21] versus the experiments of Lin *et al.* [9], which are similar to our results (Figure 4C) and similar to the results of [12]. One possible explanation is a difference in the effective concentrations of TGF- β . TGF- β has a very short half-life, and the dissolving conditions, such as carrier protein concentration, can alter the effective concentration of TGF- β . Previous authors did not report how their TGF- β was dissolved, but we found that dissolving TGF- β without carrier protein led to a steeper decline of phospho-R-Smad, similar to Massague *et al.* [11,21] (data not shown). We believe this slight discrepancy in slope is a technicality of the experiments and not fundamental to the pathway analysis.

Recent work has shown the importance of TGF- β depletion as a determinant of Smad signaling kinetics, for cells treated with low doses of TGF- β (10pM and 25pM) [47]. Our work did not emphasize low-dose contexts, but our models are consistent with observed TGF- β depletion behaviors. Figure S6 (Text S6) shows simulations of our final model, Model 8 except with lower doses of TGF- β treatment. Smad signaling was indeed dominated by TGF- β scarcity. When the Smad system was externally limited by TGF- β availability, self-limiting mechanisms and negative regulatory effects were not apparent. Negative self-regulation of the Smad system was strongly apparent in treatments with 2 ng/ml (80pM) of TGF- β , which is the dose studied in most previous experimental and computational studies of Smad dynamics.

After successfully predicting PPM1A upregulation and achieving recapitulation of the available datasets, our final contribution was to address an existing controversy about the role of proteasomal degradation in Smad signaling. We discovered that an apparent conflict about the role of degradation was in fact a mutually consistent set of trajectories that can both emerge from a single model. Degradation is intuitively understood to be a cumulative effect seen in long-term observations, but in this case the duration of observation was irrelevant, and the crucial variable for degradation was the duration of the TGF- β stimulus. MG132 (an inhibitor of proteasomal degradation) caused negligible change in pSmad2 levels (at 1,2,4,6 hr), in a system triggered with 30 min exposure to TGF-B, but MG132 caused a significant change in pSmad2 levels (at 1,2,4,6 hr), in a system triggered with long exposure to TGF-B. In other words, the importance of degradation in Smad signaling depended not on the time point at which pSmad2 was measured, but rather on the duration with which the Smad system had been induced. The consistency between the two experiments can be rationalized in retrospect because degradation depends on the area under the curve, which is large in systems with prolonged stimulus, and very small in systems with short stimulus. However, the consistency between Lin et al. and Massague et al. was not apparent prior to modeling, and mathematical inference of kinetic implications is dramatically different from the interpretations provided by the previous authors.

Computational modeling of any biochemical pathway involves several caveats and approximations, particularly when the system is as complex as Smad signaling. For our modeling of the Smad system, many interaction partners and post-translational modifications have been neglected, and some highly complex processes have been described as two-species reactions with simple mass action kinetics. Few of the rate constants have been determined from direct experiments and therefore, many parameters have been estimated by optimizing the fit between the model and the available datasets. Despite these limitations, we believe mathematical modeling provides valuable insights.

Our modeling provides a consistent, quantitative, and finegrained integration of available information about the negative regulation of phospho-R-Smad, both from published literature and from our experiments. Our combination of modeling and experiments showed that previous negative regulatory effects such as RECEPTOR DEGRADATION have a minor effect, and led us to introduce a negative feedback loop with upregulation of PPM1A. Modeling can make additional predictions (e.g., future experiments should test for peak perturbation of PPM1A binding and activity). Also, modeling has provided a new and non-obvious interpretation for the effects of MG132 treatment. When interpreting the biological meaning of observed kinetics, informal intuition can unwittingly lead to flawed conclusions. Our modeling of Smad signaling may in the future be useful to other researchers interpreting data, designing experiments, or strategizing therapeutic perturbations.

Materials and Methods

Model Specifications

1. Model structures. The reactions in our TGF- β signaling pathway model can be grouped into three sections: trafficking, Smad nucleocytoplasmic shuttling, and negative regulatory effects. Our assumptions of the receptor trafficking followed those in [25]. For Smad nucleocytoplasmic shuttling, we followed [10]. The only difference in Smad nucleocytoplasmic shuttling between [25] and [10] is that in [10], R-Smad can form a homogeneric complex. It has been shown that R-Smad can form complex with themselves [47]. Although the stoichiometry is not clear, we follow the simplest assumption in [10] that R-Smad can form homogeneric and heterogeneric complexes at the same rate. For the negative regulatory effects, we tested many possibilities based on literature findings and also our hypotheses (such as PPM1A STABILIZATION). Different effects are listed in Table 1. Here we describe each of them in details. All species names are listed in Table S1. All rate constants are listed in Table S2.

- 1) R-SMAD DEPHOSPHORYLATION was modeled as a single reaction in which nuclear phospho-R-Smad was dephosphorylated to R-Smad. The rate of dephosphorylation was proportional to the concentration of phospho-R-Smad (Reaction 32, Table S3).
- 2) RECEPTOR DEGRADATION was modeled by allowing the ligandreceptor complex in the caveolae to degrade at a rate proportional to the concentration of Smad7 (Reaction 31, Table S3).
- 3) P-SMAD DEGRADATION was modeled as a single reaction in which nuclear phospho-R-Smad was degraded at a rate proportional to its concentration (Reaction 33, Table S3), assuming that Smurf2 would be unchanged in TGF-β signaling.
- 4) ENDOGENOUS SYNTHESIS AND DEGRADATION OF R-SMAD was modeled by incorporating (a) constant production of cytoplasmic R-Smad (Reaction 34, Table S3) and (b) degradation of R-Smad (including phosphorylated and unphosphorylated forms, excluding Smad complexes), proportional to the R-Smad concentration (Reactions 35-38, Table S3), but independent of Smurf2-induced degradation.
- 5) RECEPTOR INHIBITION was modeled (in Model S1) such that I-Smad could induce degradation, inhibition, and

dephosphorylation of the ligand-receptor complex. Firstly, I-Smad (Smad7) was produced at a rate proportional to the concentration of Smad complex in the nucleus (Reactions 39-40, Table S3) and had a turnover rate proportional to its concentration (Reaction 41, Table S3). Then I-Smad could either associate with ligand receptor complex (LRC) in the caveolae (LRC_{Cave}, Reaction 43, Table S3) or could associate with LRC in the early endosome (LRC_{EE}, Reaction 44, Table S3). After association of the LRC with I-Smad, the complex could either be dephosphorylated (Reactions 45-46, Table S3) or degraded (Reactions 47-48, Table S3).

- 6) PPM1A UPREGULATION BY EXPRESSION (in Model S2) assumed that the Smad complex in the nucleus was responsible for inducing PPM1A production. That is, the rate of induced production was proportional to the concentration of Smad complex in the nucleus (Reactions 55-56, Table S3). To simulate basal (unstimulated) levels, PPM1A was also synthesized at a constant rate (Reaction 51, Table S3). All sources of PPM1A, unless bound, were degraded endogenously at a rate proportional to PPM1A concentration (Reaction 51, Table S3). Another assumption concerns the kinetics of PPM1A activity. Prior models with constant PPM1A levels used a one-step approximation for the kinetics of the dephosphorylation of phospho-R-Smad by PPM1A, but the models with explicit regulation of PPM1A employed a two-step model of catalysis (Reaction 57-60, Table S3) with reversible association/dissociation followed by irreversible catalysis.
- 7) PPM1A STABILIZATION assumed that PTEN could associate with phospho-R-Smad (Reaction 64, Table S3) and this binary complex could further associate with PPM1A to form a ternary complex pSmad2:PTEN:PPM1A (Reaction 66, Table S3). The ternary complex could dissociate in the manner it was formed, or could alternatively release the phospho-R-Smad alone and the PTEN-PPM1A as a binary complex (Reaction 67, Table S3). The PTEN-PPM1A complex was assumed to evade degradation while the unbound PPM1A would degrade spontaneously (Reactions 52-53, Table S3). PPM1A was synthesized in the cytoplasm (Reaction 52, Table S3) and was imported into the nucleus at a high rate (Reaction 54, Table S3) so that PPM1A was predominantly in the nucleus. When PTEN bound to PPM1A, PPM1A phosphatase activity was assumed to be unchanged (Reaction 61–64, Table S3). We allowed the rate of PTEN-PPM1A imported into the nucleus to differ from the rate of import for unbound PPM1A, and the actual rates were estimated numerically (Reaction 70, Table S3). PTEN in the nucleus could be exported back into the cytoplasm (Reaction 71, Table S3).

2. Kinetic parameters. 2.1 Rate constants. The list of rate constants is shown in Table S2. We have retained the values of the experimentally derived parameters cited by [25]. We have also retained previous rates for the type I and the type II receptors and the recycling rate of the ligand-receptor complex in the caveolae, which had been strongly constrained by qualitative information. For the other rate constants that were estimated by [25], we have re-estimated these parameters again in the context of our model. In particular, the rate constants for Smad nucleocytoplasmic shuttling were modified to fit our model calibration and the new findings in [10]. First of all, the concentrations of species in the nucleus are represented as their relative concentrations in the cytoplasm. For example, if the absolute concentration of Smad2 in the nucleus is, we use the relative concentration instead of in our

model to simulate the concentrations in two compartments. For example, if the absolute concentration of Smad2 in the nucleus is $[Smad2_{nuc}]^{abs}$, we use the relative concentration $[Smad2_{nuc}]^{rel} = [Smad2_{nuc}]^{abs} \cdot (V_{nuc}/V_{cyl})$ instead of $[Smad2_{nuc}]^{abs}$ in our model to simulate the concentrations in two compartments. For example, the ODEs of Smad2 using absolute concentrations are:

$$\frac{d[Smad2_{cyt}]}{dt} = -kimp_{Smad2}[Smad2_{cyt}] + kexp_{Smad2}[Smad2_{nuc}]^{abs}$$
$$\cdot \frac{V_{nuc}}{V_{cyt}} - kf_{Smad2}[Smad2_{cyt}][LRC_{EE}]$$

$$\frac{d[Smad2_{nuc}]^{abs}}{dt} = kimp_{Smad2}[Smad2_{cyl}] \cdot \frac{V_{cyt}}{V_{nuc}}$$
$$-kexp_{Smad2}[Smad2_{nuc}]^{abs}$$
$$+kdeph_{pSmad2}[pSmad2_{nuc}]^{abs}$$

After substituting $[Smad2_{nuc}]^{abs}$ with $[Smad2_{nuc}]^{rel} \cdot (V_{cyt}/V_{nuc})$:

$$\frac{d[Smad2_{cyt}]}{dt} = -kimp_{Smad2}[Smad2_{cyt}] + kexp_{Smad2}[Smad2_{nuc}]^{rel} - kf_{Smad2}[Smad2_{cyt}][LRC_{EE}]$$

$$\frac{d[Smad2_{nuc}]^{rel}}{dt} = kimp_{Smad2}[Smad2_{cyt}] - kexp_{Smad2}[Smad2_{nuc}]^{rel} + kdeph_{pSmad2}^{rel}[pSmad2_{nuc}]^{rel}$$

where

$$kdeph_{pSmad2}^{rel} = kdeph_{pSmad2} \cdot \frac{V_{cyt}}{V_{nuc}} = 3 \cdot kdeph_{pSmad2}$$

The export rates computed in [6] are identical to our $kexp_{Smad2}$ and $kexp_{Smad4}$. However, the import in [6] are actually $kimp'_{Smad2} = kimp_{Smad2} \cdot (V_{cyt}/V_{nuc})$ and $kimp'_{Smad4} = kimp_{Smad4} \cdot (V_{cyt}/V_{nuc})$. So our import rates are $kimp_{Smad2} = kimp'_{Smad4} \cdot (V_{nuc}/V_{cyt}) = 0.0027/3 = 0.0009s^{-1} = 0.054 \text{ min}^{-1}$ and $kimp_{Smad4} = kimp'_{Smad4} \cdot (V_{nuc}/V_{cyt}) = 0.0027/3 = 0.0009s^{-1} = 0.054 \text{ min}^{-1}$. Note that the rates of the reactions in the nucleus (e.g. $kdeph_{pSmad2}$) are substituted because the concentrations of the reactants are relative. But all these rates are estimated so we do not need to substitute any values of them.

Secondly, it was found that the import rate of Smad complex is higher than the monomeric Smad2 [10]. Therefore, we set the import and export rate of Smad complex according to [10].

2.2 Initial concentrations. The initial concentrations of the receptors followed those in [25], as the rate constants for receptor trafficking were kept the same as those in [25]. For Smad2 and Smad4, we can derive their equilibrium concentrations based on their shuttling rates and total amount. At steady state,

$$\begin{cases} \frac{[Smad2_{cyt]}}{[Smad2_{nuc}]^{rel}} = \frac{kexp_{Smad2}}{kimp_{Smad2}} = \frac{0.348min^{-1}}{0.054min^{-1}} = 6.444 \\ [Smad2_{nuc}]^{rel} = [Smad2_{total}]^{rel} \end{cases}$$

$$\begin{cases} \frac{[Smad4_{cyt}]}{[Smad4_{nuc}]^{rel}} = \frac{kexp_{Smad4}}{kimp_{Smad4}} = \frac{0.054min^{-1}}{0.054min^{-1}} = 1.000\\ [Smad4_{cyt}] + [Smad4_{nuc}]^{rel} = [Smad4_{total}]^{rel} \end{cases}$$

The relative concentrations of total Smad2 and total Smad4 in HaCaT cells are 571.43 nM and 1333.33 nM [12,25].

For each model, all initial concentrations were set to their equilibrium points (i.e., their steady-state levels) in unstimulated cells, computed by simulating the system for 10^4 minutes (1 week) without TGF- β . The initial concentrations are shown in Table S1.

Note that in the absence of TGF- β stimulation, only a small number of reactions occur, mainly production, degradation, and trafficking. The rate parameters for these reactions were the same for all models (Tables S2 and S5), causing all models to have the same steady-state concentrations, for all species they share in common. Note that some species are specific to certain models, such as cytosolic PPM1A appearing in (Model 8 and S3). This modification does not affect the steady-state levels of the other species, because the reactions catalyzed by PPM1A do not occur until after TGF- β stimulation.

2.3 Model simulation and parameter estimation. The model simulation and parameter estimation were performed using KroneckerBio toolbox in Matlab. The KroneckerBio toolbox basically calls the ode15 s function in Matlab to solve the system of ODE equations and the finincon function in Matlab to estimate parameters. Multiple initial guesses were generated randomly in order to achieve a more global optimum in parameter estimations. Sum of squared errors were used as the objective function to optimize the model to experimental data.

The parameters related to each negative regulatory effect are listed in Table S4. The best-fit parameters of each model are listed in Table S5.

Biological Assays

1. HaCaT cell culture and TGF-\beta treatment. HaCaT cells (from Cell Lines Service) were cultured following the protocol provided by the manufacturer. DMEM culture medium with 10% FBS was used to culture the cells. DMEM culture medium without FBS was used during treatment of TGF- β .

2. ELISA for phosphorylated Smad2 and total Smad2. ELISA kits (from Cell Signaling) were used to quantify phosphorylated Smad2 and total Smad2. Whole cell lysates were collected using attached cell lysis buffer and following the cell lysis protocol in the kits. Sample dilutions for phosphorylated Smad2 and total Smad2 are 1 time and 100 times respectively. Serial dilutions of one sample were measured to check the linear range of the readouts.

3. Western blot for total type I receptor and PPM1A. Whole cell lysates were collected using RIPA buffer from HaCaT cells for western blots. Antibodies against the type I receptor (Santa Cruz) and PPM1A (Abcam) were used following manufacturers' supplier's instructions. Primary antibody dilutions of 1:7500 and 1:250 were used for the type I receptor and PPM1A respectively. The quantification of the band intensities was preformed using ImageJ.

Supporting Information

Figure S1 Analysis of receptor internalization in Model 1. (**A**) Perturbation analysis of the rate of ligand-receptor complex (LRC) internalization. Log Parameter Perturbation is the log ratio of perturbed LRC internalization rate (the rate for internalizing early endosome and caveolae were changed with the same ratio) to its original value. Blue curve shows the result of perturbation analysis in Model 1, while red curve shows the result of perturbation analysis after changing the production rates of T1R (10-fold decrease) and T2R (10-fold increase) in Model 1. (**B**) Inhibition of LRC internalization in Model 1 after changing the production rates of T1R (10-fold decrease) and T2R (10-fold increase). Curves change from blue to red as LRC internalization rate decreases from 1 to 10^{-3} in log scale. (**C**) Adaptation Index change (x-axis) with 10-fold decrease of each parameter. (**D**) Adaptation Index change (x-axis) with 10-fold decrease of each parameter. (**E**) Dose response at 45 min. (**F**) Dose response at 45 min. In panel E and F, blue curve shows dose response in Model 1, while red curve shows dose response in Model 1 after changing the production rates of T1R (10-fold decrease) and T2R (10-fid increase) in Model 1.

(EPS)

Figure S2 Fitted models with extended receptor inhibition effects (Model S1). Each blue point represents a single model. The x-axis is the type I receptor (T1R) level simulated at 24 hr. The y-axis is the ratio of I-Smad-bound ligand-receptor complex in the early endosome and caveolae. We rescaled the axes to better visualize the majorities of the data points (upper left sub-figure). Models in the red box region should be able to explain both the type I receptor level and localization of I-Smad. However, no fitted model falls in the red box. (EPS)

Figure S3 Sensitivity analysis heat map of Model 5. (**A**) The sensitivity of the P-Smad degradation rate to each species with relative perturbations of the rate from 10^{-4} to 10^2 . (**B**) The sensitivity of the receptor degradation rate to each species with relative perturbations of the rate from 10^{-4} to 10^2 . (**E**PS)

Figure S4 Model predictions PPM1A upregulation. (**A**) Model S2, in which PPM1A is upregulated by Smad complex in the nucleus, was fitted to the long-exposure and the short-exposure phospho-R-Smad experimental data. (**B**) Model S2 predicted PPM1A upregulation under long-exposure of TGF- β (green curve). (**C**) Model S2 predicted unchanged T1R levels (green curve), in agreement with our experimental results (green dots). (**D**) Model S2 predicted unchanged total R-Smad levels (green curve), in agreement with our experimental results (green dots). (EPS)

Figure S5 Model-based predictions of PPM1A STABILIZATION dynamics. (**A**) Model S3, in which PPM1A is stabilized by PTEN, was fitted to the long-exposure and the short-exposure phospho-R-Smad experimental data. (**B**) Model S3 predicted early PPM1A upregulation (within 1hr) under long-exposure treatments with TGF- β (green curve). This is in agreement with our experimental measurements of PPM1A (greed dots). (**C**) Model S3 predicted unchanged T1R levels (green curve), in agreement with our experimental results (green dots). (**D**) Model S3 predicted unchanged total R-Smad levels (green curve), in agreement with our experimental results (green dots). (EPS)

Figure S6 TGF- β dose response. (**A**) Simulated P-Smad2 levels at 45 min under different doses of TGF- β treatment. (**B**) Simulated P-Smad2 levels at 24 hr under different doses of TGF- β treatment. (**C**) Simulations of the P-Smad2 dynamics with different doses of TGF- β . The color of the curve turns from blue to red as TGF- β dose increases (0.025, 0.0625, 0.125, 0.25, 0.5, 1, 2 ng/ml). (EPS)

Figure S7 Sum of Squared Errors (SSE) of the best-fit models. SSE was plotted separately for P-Smad2 and PPM1A in Model 8, S2 and

S3 so that the goodness-of-fit of these models can be compared to other models.

(EPS)

Figure S8 Modeling cycloheximide treatment. (**A**) Dataset originated from [49] and quantified from [50]. The blue dots show P-Smad2 levels in the nucleus in HaCaT cells after treated with 2 ng/ml of TGF- β . The red dots show P-Smad2 levels under the same condition but HaCaT cells were pretreated with cycloheximide for 30 min. (**B**) Simulated P-Smad2 levels from Model 8 without cycloheximide treatment (blue curve) and with cycloheximide treatment (all production rates are half of their original values, red curve). (EPS)

Figure S9 The dynamics of the PTEN-PPM1A complex in Model 8. In this model, we assumed PTEN is initially cytoplasmic, and we assumed that phospho-R-Smad could induce association between PTEN and PPM1A. The simulation shows that TGF- β stimulation would rapidly induce the formation of a cytosolic PTEN-PPM1A complex (blue curve), which would then translocate to the nucleus. The speed of nuclear translocation would affect the functional availability of PPM1A toward phospho-R-Smad in the nucleus. If PTEN binding slows the nuclear translocation of PPM1A (as simulated in Model 8), then PTEN-induced stabilization could have a complex effect of causing a short-term delay in PPM1A availability even though it causes a long-term increase in PPM1A abundance. Note that TGF- β does not alter the total expression of PTEN (green curve) in this model. (EPS)

Table S1Initial Concentrations (I.C.) in nM.(PDF)

Table S2Rate constants.(PDF)

Table S3 Reactions table: All reactions in Models1–8 and Models S1–S3 with rate constants labeled. (PDF)

Table S4Table of negative regulatory effects and their relatedrate constants.

(PDF)

Table S5Table of estimated parameters in Models 1–8 andModels S2–S3.

(PDF)

 $Text\ S1$ The rate of receptor internalization does not affect the peak and decline of phospho-R-Smad when TGF- β is saturating.

(PDF)

Text S2 Extended I-Smad-mediated receptor inhibition. (PDF)

Text S3 Sensitivity analysis.

(PDF)

Text S4 PPM1A upregulation could possibly be a slow-mode effects.

(PDF)

Text S5 Model-based prediction of PPM1A STABILIZATION dynamics. (PDF)

 $Text\ S6$ Transient and sustained signaling in the TGF- β signaling pathway.

(PDF)

Text S7 Evaluation of best-fits of all models. (PDF)

Text S8 Modeling cycloheximide treatment. (PDF)

Dataset S1 SBML file for Model 1. (XML)

Dataset S2 SBML file for Model 2. (XML)

Dataset S3 SBML file for Model 3. (XML)

Dataset S4 SBML file for Model 4. (XML)

Dataset S5 SBML file for Model 5. (XML)

Dataset S6 SBML file for Model 6. (XML)

Dataset S7 SBML file for Model 7. (XML)

References

- Yingling JM, Blanchard KL, Sawyer JS (2004) Development of TGF-beta signalling inhibitors for cancer therapy. Nature reviews Drug discovery 3: 1011-1022.
- Massague J (1998) TGF-beta signal transduction. Annu Rev Biochem 67: 753– 791.
- Shi Y, Massague J (2003) Mechanisms of TGF-beta signaling from cell membrane to the nucleus. Cell 113: 685–700.
- Attisano L, Wrana JL (2002) Signal transduction by the TGF-beta superfamily. Science 296: 1646–1647.
- Bassing CH, Yingling JM, Howe DJ, Wang T, He WW, et al. (1994) A transforming growth factor beta type I receptor that signals to activate gene expression. Science 263: 87–89.
- Schmierer B, Hill CS (2005) Kinetic analysis of Smad nucleocytoplasmic shuttling reveals a mechanism for transforming growth factor beta-dependent nuclear accumulation of Smads. Mol Cell Biol 25: 9845–9858.
- Massague J, Wotton D (2000) Transcriptional control by the TGF-beta/Smad signaling system. EMBO J 19: 1745–1754.
- Massague J, Blain SW, Lo RS (2000) TGFbeta signaling in growth control, cancer, and heritable disorders. Cell 103: 295–309.
- Lin X, Duan X, Liang YY, Su Y, Wrighton KH, et al. (2006) PPM1A functions as a Smad phosphatase to terminate TGFbeta signaling. Cell 125: 915–928.
- Schmierer B, Tournier AL, Bates PA, Hill CS (2008) Mathematical modeling identifies Smad nucleocytoplasmic shuttling as a dynamic signal-interpreting system. Proc Natl Acad Sci U S A 105: 6608–6613.
- Lo RS, Massague J (1999) Ubiquitin-dependent degradation of TGF-betaactivated smad2. Nat Cell Biol 1: 472–478.
- Zi Z, Feng Z, Chapnick DA, Dahl M, Deng D, et al. (2011) Quantitative analysis of transient and sustained transforming growth factor-beta signaling dynamics. Molecular systems biology 7: 492.
- Afrakhte M, Moren A, Jossan S, Itoh S, Sampath K, et al. (1998) Induction of inhibitory Smad6 and Smad7 mRNA by TGF-beta family members. Biochem Biophys Res Commun 249: 505–511.
- Itoh S, Landstrom M, Hermansson A, Itoh F, Heldin CH, et al. (1998) Transforming growth factor betal induces nuclear export of inhibitory Smad7. J Biol Chem 273: 29195–29201.
- Ebisawa T, Fukuchi M, Murakami G, Chiba T, Tanaka K, et al. (2001) Smurfl interacts with transforming growth factor-beta type I receptor through Smad7 and induces receptor degradation. J Biol Chem 276: 12477–12480.
- Hayashi H, Abdollah S, Qiu Y, Cai J, Xu YY, et al. (1997) The MAD-related protein Smad7 associates with the TGFbeta receptor and functions as an antagonist of TGFbeta signaling. Cell 89: 1165–1173.
- Kavsak P, Rasmussen RK, Causing CG, Bonni S, Zhu H, et al. (2000) Smad7 binds to Smurf2 to form an E3 ubiquitin ligase that targets the TGF beta receptor for degradation. Mol Cell 6: 1365–1375.
- Zhang L, Zhou F, Drabsch Y, Gao R, Snaar-Jagalska BE, et al. (2012) USP4 is regulated by AKT phosphorylation and directly deubiquitylates TGF-beta type I receptor. Nat Cell Biol 14: 717–726.
- Eichhorn PJ, Rodon L, Gonzalez-Junca A, Dirac A, Gili M, et al. (2012) USP15 stabilizes TGF-beta receptor I and promotes oncogenesis through the activation of TGF-beta signaling in glioblastoma. Nat Med 18: 429–435.

Dataset S8 SBML file for Model 8. (XML)

Dataset S9 SBML file for Model S2. (XML)

Dataset \$10 SBML file for Model S3. (XML)

Acknowledgments

Thanks to Bo Kim (Massachusetts Institute of Technology) for assistance with KroneckerBio toolbox, and to Josh Apgar and Jared Toettcher (both from Massachusetts Institute of Technology) for guidance of KroneckerBio toolbox. Thanks to Prof. Bruce Tidor (Massachusetts Institute of Technology) for distributing KronekcerBio toolbox. Thanks to Ser-Mien Chia (National University of Singapore) and Lakshmi Venkatraman (Singapore-MIT Alliance) for helpful discussions.

Author Contributions

Conceived and designed the experiments: JW LTK HY. Performed the experiments: JW ICN RJ. Analyzed the data: JW LTK. Wrote the paper: JW LTK HY JKW PST.

- Lin X, Liang M, Feng XH (2000) Smurf2 is a ubiquitin E3 ligase mediating proteasome-dependent degradation of Smad2 in transforming growth factorbeta signaling. J Biol Chem 275: 36818–36822.
- Alarcon C, Zaromytidou AI, Xi Q, Gao S, Yu J, et al. (2009) Nuclear CDKs drive Smad transcriptional activation and turnover in BMP and TGF-beta pathways. Cell 139: 757–769.
- Inui M, Manfrin A, Mamidi A, Martello G, Morsut L, et al. (2011) USP15 is a deubiquitylating enzyme for receptor-activated SMADs. Nat Cell Biol 13: 1368– 1375.
- Aggarwal K, Massague J (2012) Ubiquitin removal in the TGF-beta pathway. Nat Cell Biol 14: 656–657.
- Vilar JM, Jansen R, Sander C (2006) Signal processing in the TGF-beta superfamily ligand-receptor network. PLoS Comput Biol 2: e3.
- Klipp E, Zi Z (2007) Constraint-based modeling and kinetic analysis of the smad dependent tgf-Beta signaling pathway. PLoS ONE 2: e936.
- Melke P, Jonsson H, Pardali E, ten Dijke P, Peterson C (2006) A rate equation approach to elucidate the kinetics and robustness of the TGF-beta pathway. Biophys J 91: 4368–4380.
- Chung SW, Miles FL, Sikes RA, Cooper CR, Farach-Carson MC, et al. (2009) Quantitative modeling and analysis of the transforming growth factor beta signaling pathway. Biophys J 96: 1733–1750.
- Bu S, Kapanadze B, Hsu T, Trojanowska M (2008) Opposite effects of dihydrosphingosine 1-phosphate and sphingosine 1-phosphate on transforming growth factor-beta/Smad signaling are mediated through the PTEN/PPM1Adependent pathway. J Biol Chem 283: 19593–19602.
- Hjelmeland AB, Hjelmeland MD, Shi Q, Hart JL, Bigner DD, et al. (2005) Loss of phosphatase and tensin homologue increases transforming growth factor betamediated invasion with enhanced SMAD3 transcriptional activity. Cancer research 65: 11276–11281.
- Toettcher JE, Apgar JF, Castillo AR, Tidor B, White J (2011) Recycling Circuit Simulation Techniques for Mass-Action Biochemical Kinetics. In: Li P, Silveira LM, Feldmann P, eds., editors. chapter in Simulation and Verification of Electronic and Biological Systems: Springer.
- Edlund S, Lee SY, Grimsby S, Zhang S, Aspenstrom P, et al. (2005) Interaction between Smad7 and beta-catenin: importance for transforming growth factor beta-induced apoptosis. Mol Cell Biol 25: 1475–1488.
- Shi W, Sun C, He B, Xiong W, Shi X, et al. (2004) GADD34-PP1c recruited by Smad7 dephosphorylates TGFbeta type I receptor. J Cell Biol 164: 291–300.
- Li L, Xin H, Xu X, Huang M, Zhang X, et al. (2004) CHIP mediates degradation of Smad proteins and potentially regulates Smad-induced transcription. Mol Cell Biol 24: 856–864.
- Wan M, Tang Y, Tytler EM, Lu C, Jin B, et al. (2004) Smad4 protein stability is regulated by ubiquitin ligase SCF beta-TrCP1. J Biol Chem 279: 14484–14487.
- Lin F, Morrison JM, Wu W, Worman HJ (2005) MAN1, an integral protein of the inner nuclear membrane, binds Smad2 and Smad3 and antagonizes transforming growth factor-beta signaling. Hum Mol Genet 14: 437–445.
- 36. Pan D, Estevez-Salmeron LD, Stroschein SL, Zhu X, He J, et al. (2005) The integral inner nuclear membrane protein MAN1 physically interacts with the R-Smad proteins to repress signaling by the transforming growth factor-{beta} superfamily of cytokines. J Biol Chem 280: 15992–16001.

- Remy I, Montmarquette A, Michnick SW (2004) PKB/Akt modulates TGFbeta signalling through a direct interaction with Smad3. Nat Cell Biol 6: 358– 365.
- Conery AR, Cao Y, Thompson EA, Townsend CM, Jr., Ko TC, et al. (2004) Akt interacts directly with Smad3 to regulate the sensitivity to TGF-beta induced apoptosis. Nat Cell Biol 6: 366–372.
- Song K, Wang H, Krebs TL, Danielpour D (2006) Novel roles of Akt and mTOR in suppressing TGF-beta/ALK5-mediated Smad3 activation. EMBO J 25: 58–69.
- Gomis RR, Alarcon C, Nadal C, Van Poznak C, Massague J (2006) C/EBPbeta at the core of the TGFbeta cytostatic response and its evasion in metastatic breast cancer cells. Cancer Cell 10: 203–214.
- Zhu Q, Krakowski AR, Dunham EE, Wang L, Bandyopadhyay A, et al. (2007) Dual role of SnoN in mammalian tumorigenesis. Mol Cell Biol 27: 324–339.
- Krakowski AR, Laboureau J, Mauviel A, Bissell MJ, Luo K (2005) Cytoplasmic SnoN in normal tissues and nonmalignant cells antagonizes TGF-beta signaling by sequestration of the Smad proteins. Proc Natl Acad Sci U S A 102: 12437– 12442.
- Levy L, Howell M, Das D, Harkin S, Episkopou V, et al. (2007) Arkadia activates Smad3/Smad4-dependent transcription by triggering signal-induced SnoN degradation. Mol Cell Biol 27: 6068–6083.

- Bakin AV, Tomlinson AK, Bhowmick NA, Moses HL, Arteaga CL (2000) Phosphatidylinositol 3-kinase function is required for transforming growth factor beta-mediated epithelial to mesenchymal transition and cell migration. J Biol Chem 275: 36803–36810.
- Goumans MJ, Valdimarsdottir G, Itoh S, Rosendahl A, Sideras P, et al. (2002) Balancing the activation state of the endothelium via two distinct TGF-beta type I receptors. EMBO J 21: 1743–1753.
- Chen G, Ghosh P, Longo DL (2011) Distinctive mechanism for sustained TGFbeta signaling and growth inhibition: MEK1 activation-dependent stabilization of type II TGF-beta receptors. Mol Cancer Res 9: 78–89.
- Clarke DC, Brown ML, Erickson RA, Shi Y, Liu X (2009) Transforming growth factor beta depletion is the primary determinant of Smad signaling kinetics. Molecular and cellular biology 29: 2443–2455.
- Nakao A, Afrakhte M, Moren A, Nakayama T, Christian JL, et al. (1997) Identification of Smad7, a TGFbeta-inducible antagonist of TGF-beta signalling. Nature 389: 631–635.
- Inman GJ, Nicolas FJ, Hill CS (2002) Nucleocytoplasmic shuttling of Smads 2, 3, and 4 permits sensing of TGF-beta receptor activity. Mol Cell 10: 283–294.
- Vilar JM, Saiz L (2011) Trafficking coordinate description of intracellular transport control of signaling networks. Biophys J 101: 2315–2323.